Author’s response to reviews

Title: CCL4 Is the Only Predictor for Non-Responder in GT-1 CHC Patients with Favorable IL28B genotype Treated with PegIFN/RBV

Authors:

Chia-Chen Lin (lydia05025137@hotmail.com)
Chun-Yen Lin (chunyenlin@gmail.com)
I-Shyan Sheen (happy95kevin@gmail.com)
Shih-Huan Su (franktw123@hotmail.com)
Wen-Juei Jeng (rachel.jeng@gmail.com)
Chien-Hao Huang (huangchianhou@gmail.com)
Wei Teng (b101090023@gmail.com)
Wei-Ting Chen (weiting1972@gmail.com)
Yi-Cheng Chen (yichengliver@gmail.com)

Version: 2 Date: 01 Dec 2017

Author’s response to reviews:

Dear Dr. Gritti
Editorial Board Member,
BMC Gastroenterology

Thank you for reviewing our revised manuscript entitled "CCL4 Is the Only Predictor for Non-Responder in GT-1 CHC Patients with Favorable IL28B genotype Treated with PegIFN/RBV".

We provide in this letter concise point-by-point responses to your comments, along with a revised manuscript for consideration in your journal.

1. We had amended the heading 'Introduction' to 'Background'.
2. We had added a Conclusions section after the discussion.
3. We had included a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.
4. We had put all the Figure Legends together, after the References, under the heading 'Figure Legends'.

5. We had uploaded a single, final, clean version of manuscript.

Sincerely,

Chia-Chen Lin, M.D.

Chun-Yen Lin, M.D./D.Phil.